

ZBW

A Service of

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Dickie, Paul; Fan, Emma Xiaogin

Working Paper

Banks and Corporate Debt Market Development

ERD Working Paper Series, No. 67

Provided in Cooperation with:

Asian Development Bank (ADB), Manila

Suggested Citation: Dickie, Paul; Fan, Emma Xiaoqin (2005): Banks and Corporate Debt Market Development, ERD Working Paper Series, No. 67, Asian Development Bank (ADB), Manila, https://hdl.handle.net/11540/1897

This Version is available at: https://hdl.handle.net/10419/109264

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



http://creativecommons.org/licenses/by/3.0/igo

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ADB

ERD Working Paper ECONOMICS AND RESEARCH DEPARTMENT SERIES

Banks and Corporate

Debt Market Development

Paul Dickie and Emma Xiaoqin Fan

April 2005

Asian Development Bank

ERD Working Paper No. 67

BANKS AND CORPORATE DEBT MARKET DEVELOPMENT

PAUL DICKIE AND EMMA XIAOQIN FAN

April 2005

Paul Dickie is an Associate Professor of International Finance at the Victoria University of Wellington, New Zealand; Emma Xiaoqin Fan is an Economist in the Economics and Research Department, Asian Development Bank. The helpful comments of Anella Munro and Akiko Terada-Hagiwara, the excellent research assistance of Cheng Fude and Gmelina Guiang, and support of the Asia 2000 Foundation are gratefully acknowledged.

- -

Asian Development Bank 6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines www.adb.org/economics

©2005 by Asian Development Bank April 2005 ISSN 1655-5252

The views expressed in this paper are those of the author(s) and do not necessarily reflect the views or policies of the Asian Development Bank.

FOREWORD

The ERD Working Paper Series is a forum for ongoing and recently completed research and policy studies undertaken in the Asian Development Bank or on its behalf. The Series is a quick-disseminating, informal publication meant to stimulate discussion and elicit feedback. Papers published under this Series could subsequently be revised for publication as articles in professional journals or chapters in books.

CONTENTS

Abstra	act	vii
I.	Introduction	1
II.	Relationship between Banks and Bond Markets	2
III.	Hypothesis and Model Specification	4
	A. PlayersB. InstitutionsC. Economic Interrelationships and Incentives	4 6 7
IV.	Data and Methodology	8
٧.	Results	11
VI.	Equity Market Development	13
VII.	Conclusions and Policy Implications	14
	References	16

ABSTRACT

This paper explores the factors associated with the development of corporate debt markets using panel data covering 30 countries from 1989 to 2002. The results support Rajan and Zingales's (2003) "interest group" theory of financial development that banks appear to oppose corporate debt market development as a potential force for their own disintermediation. The more concentrated the banking sector, the smaller the corporate bond market relative to the size of the economy. There is also evidence that the opening up of cross-border merger and acquisition activities and the presence of global corporations seem to weaken the influence of domestic banks. While outward-looking economic policies can reduce the power of domestic banks, the major countervailing force appears to be that committed governments recognizing corporate debt markets can enhance the resilience of their domestic economies.

I. INTRODUCTION

Domestic bond markets are an important resource for firms seeking access to long-term local currency debt to reduce maturity and currency mismatches. This importance was highlighted by the fragility of economies during the Asian financial crisis. A widely accepted lesson to emerge from the crisis was that borrowers can face severe problems when they rely on short-term foreign currency funding. The lack of local currency debt markets was recognized as at least part of the reason why firms undertook short-term borrowing in international markets (Dornbusch 2002, Tornell 2002, Knight 1998, Summers 1999, Stone 2000). This inability to borrow long-term in domestic currency has been characterized as the "original sin" (Eichengreen and Hausmann 1999; Hausmann, Panizza, and Stein 2001).

Well-developed corporate debt markets have increasingly been seen as an integral part of resilient financial systems. Bank for International Settlements (BIS) statistics indicate that 80 percent of the debt securities outstanding in domestic debt markets, and 78 percent within international debt markets, had remaining maturities of more than one year at the end of 2000. By comparison, the maturity of international bank loans is predominantly short-term in nature, with only 34 percent of the outstanding loans being for a term of over one year. Given the widespread adoption of variable rate loans in the international market that allow longer duration loans without substantial interest rate risk, it is likely that the domestic bank loan markets have an even lower percentage of longer-term maturities. Clearly debt markets offer a spectrum of longer-term debt maturities.

Banks' reluctance to provide long-term credits to corporations could be explained by an understandable hesitancy to widen the maturity mismatch on their own balance sheet, as banks' funding is generally in the form of short-term deposits. While banks can partially meet corporate requirements for longer-term loans by rolling over short-term credits, this solution leaves projects with unmatched asset and liability maturities.

Since the Asian crisis, a growing number of emerging market economies have made efforts to develop local bond markets. Yet, at present, corporate debt markets still appear to be underdeveloped around the world. Less than one third of the over 100 countries with equity markets have corporate debt markets. Even where they exist, corporate debt markets average only one tenth the size of the corresponding equity markets. Most emerging market domestic corporate debt securities are plagued by a lack of issuance in primary markets and liquidity in secondary markets. This is in marked contrast to early capital market development in Europe and the United States (US) when debt markets dominated (Ferguson 2001, Sylla 1995).

Questions naturally arise as to what accounts for this lag in the development of local currency debt markets, and what lessons do previous experiences hold for policymakers seeking to create environments where bond markets can flourish. In an initial attempt to identify factors associated with debt market development, this paper examines the determinants for the development of corporate debt securities markets using BIS data from 1989 to 2002 for more than 30 countries.

This paper is particularly interested in exploring the relationship between banking sector structure and corporate bond market development. The central question the paper seeks to investigate is whether a highly concentrated banking sector impedes bond market development. The need to study the relationship is underpinned by the potential clash brought about by the fact that banks are active participants in corporate bond market activities, yet also potential opponents to bond market development due to the threat of disintermediation it may pose to the banking sector. A better understanding of the bank-corporate bond market relationship, along with other determinants of bond market development, can shed light on how best to provide appropriate policies to facilitate bond market development.

The results of this paper generally support the "interest group" theory proposed by Rajan and Zingales (2003) that emphasizes incumbents' opposition to financial development because of the increased competition it introduces. It provides some evidence that banks, as the principal incumbents, have the tendency to oppose the development of corporate debt markets, given that debt market development can entail disintermediation for banks. The ability of banks to oppose bond market development is related to the structure of the banking sector. A more concentrated banking sector possesses more capacity to deter bond market development. The paper also provides some evidence that the opening up of cross-border merger and acquisition (M&A) activities and the presence of large global corporations seem to weaken the influence of domestic banks. But the most important countervailing influence is that of governments that support corporate debt market development.

Section II of this paper discusses the relationship between banks and corporate debt market proposed in the existing literature. Section III explores possible influences on debt market development with a view to model specifications for the empirical analysis. Section IV describes the data and the methodology employed. Section V presents the results of the analysis, Section VI provides a brief examination of determinants of equity markets as a comparison of debt markets, and Section VII concludes with some policy implications.

II. RELATIONSHIP BETWEEN BANKS AND BOND MARKETS

An important aspect of the development of bond markets is their impact on the banking system. Do bond markets take good lending business away from banks? If so, the quality of bank loans may deteriorate as banks lose better borrowers so that banks will naturally oppose bond market development. However, few theoretical and empirical studies have explored this question.

A recent study by Rajan and Zingales (2003) sheds some light on this question. Their interest group theory states that incumbents oppose financial development because it breeds competition. Because of their privileged access to finance in less developed financial systems, incumbents enjoy a positional rent that will be impaired by financial development. Permitting new competitors to enter can draw away profits, while at the same time the incumbent's old skills also become redundant as new credit evaluation and risk management become necessary. Thus, financial development introduces competition, destroys the existing financial institution's rents and relations, and renders the incumbents' human capital obsolete. All these lead to lower profits for incumbent financial institutions. A central

assumption of this argument is the belief that financial development aids the entry of new firms, thus enhancing competition. Naturally, the incumbents' opposition to financial development will be stronger if they have more market power.

This theory suggests that banks with market power may attempt to stifle the development of corporate bond markets. This is because the development of corporate security markets may compete for good borrowers from banks, the quality of bank loans may drop, the task of risk management may become harder as loans become riskier, and existing human capital may need to be updated. Banks with strong market power will have a stronger tendency and larger capacity to oppose corporate bond market development.

While very few empirical studies directly explore the relationship between the banking sector structure and corporate debt market development, some studies do touch upon bank-corporate bond market relationships. Hawkins (2002) observed that highly rated companies issue more bonds than do lower rated companies. Diamond (1994) has noted that bank loans must bear higher intermediation costs associated with banks' branch networks and required capital. With the relative cost advantages of debt securities, corporations that seek cost savings and have good reputations to issue debt securities will tend to do so when a debt market is available. Also as outlined by Bolton and Freixas (2000), less financially secure firms prefer bank loans due to the greater flexibility in rescheduling, while the larger, creditworthy firms seek to tap the bond markets. Analyzing the determinants of the financing choice for a sample of 1,560 new debt financing undertakings by publicly traded firms, Denis and Mihov (2003) found that the choice of debt instrument is strongly linked to the credit history and current credit quality of the issuing firm. Firms with high credit quality exhibit a strong preference for public debt, while firms with credit ratings toward the middle of the spectrum borrow from banks, and those at the bottom of the credit rating spectrum borrow from nonbank private sources. These studies suggest that corporate bond markets compete with banks and can draw away good borrowers from banks.

The relationship between banks and corporate bond market developments are not unambiguous, however. Demirguc-Kunt and Huizinga (2001), using bank level data for a large number of developed and developing countries, show that for countries with underdeveloped financial systems, a move toward a more developed financial system reduces bank profitability and margins. However, controlling for both bank and market developments, the financial structure per se does not have an independent effect on bank performance. An empirical test conducted by Jiang and Law (2001) found bond issuance and bank lending are positively correlated both in countries of the Organisation for Economic Cooperation and Development and the emerging economies. Eichengreen and Luengnaruemitchai (2004) examined the relationship between the banking sector and bond market development along with other factors using information from 41 countries from BIS reports. They found that countries with competitive, well-capitalized banking systems have larger bond markets. These studies point to a complementary relationship between banks and bond market development.

The mixed results suggest a need for further research to explore the relationship between banks and corporate bond market development. This is especially so in light of the limited empirical literature on the relationship between banks and corporate debt market development in particular, and on the determinants of corporate debt market development in general. This study seeks to make some empirical attempt in this regard.

III. HYPOTHESIS AND MODEL SPECIFICATION

The major relationship this paper concerns itself with is that between the banking sector structure and corporate bond market development. Based on interest group theory, we postulate a negative relationship between banking sector concentration and corporate bond market development. In measuring bond market development, we focus on an important variable: the size of the corporate bond market relative to the size of an economy. Size is not the only indication of bond market development, however. Other factors such as efficiency and liquidity are also important. Nevertheless, size is probably the most important indictor measuring bond market development. The major hypothesis is that the more concentrated the banking sector, the more power that banks can exercise, and the less likely it will be that corporate debt markets will grow.

The relationship between banks and corporate bond markets is examined along with other possible determinants of corporate bond market development discussed below.¹

A. Players

1. Global Corporations (+)

It is generally observed that more creditworthy firms are more likely to tap bond markets. Such corporations can issue securities on a "name" or reputation basis to attract a high level of acceptance from investors. A viable group of issuers and investors will help to create market liquidity in both the primary and secondary markets. Due to network effects, the benefits to corporate bond investors will rise with the number of issuers (Gandal 2002). The market power of banks may thus be rivaled by global corporations with a high degree of name acceptance that have potential access to the debt markets.

However, even large and well-established corporations need to have an unassailable position to ensure a continued flow of banking services while undertaking a shift in financing from banks to the market. For example, it is normal practice to have a back-up line of credit from a bank for a commercial paper program in case the debt markets prove to be inaccessible when corporate credit is required. As such, a corporation undertaking a shift to a debt market still needs to have the cooperation of a bank. It is therefore unlikely that the role of banks will be totally offset.

2. Government Commitments (+)

Due to problems related to networks, asymmetric information, as well as the duopolistic structure of the banking systems, financial development often requires the government to play a role.

In the initial stage of bond market development, the government may need to take the lead in getting the process under way by bringing together key market players. Governments many also need to ensure that corporate bond markets are not held hostage by agencies such as commercial banks, central banks, and regulatory agencies such as securities and exchange commissions. Rajan

¹ (+) indicates a positive relationship between a particular variable and bond market development, (-) refers to a negative relationship, and (?) denotes an undefined relationship.

and Zingales (2003) note the negative influence of commercial banks in Japan's post-World War II period when, with the support of the government, they colluded to sharply reduce corporations' access to debt markets.

The importance of governments is demonstrated by the experience of Republic of Korea (Korea) and Malaysia. Governments in these two countries made bond market development a priority and took steps to facilitate this. In Malaysia, the government directed the utilization of funds from the Malaysian Provident Fund by infrastructure providers. It created a national mortgage corporation (Camagas Berhad) in 1988 to securitize banks mortgage loans and develop the private bond market. Camagas bonds enjoyed favorable regulatory treatment, including being treated as liquid assets for the purpose of bank investments. The scheme acted as a catalyst for corporate debt market development. The Employee Provident Fund was actively encouraged to invest in corporate bonds to help finance the development of infrastructure and energy investments of domestic corporations. About half of the corporate bonds was guaranteed by banks to improve their market acceptability (Dalla 1995, Shirai 2001). In addition, the government promoted the development of the needed infrastructure for bond market development including bond rating agencies and made their ratings mandatory. By 2000, the corporate debt market amounted to 47 percent of GDP, having increased from 4 percent in 1989.

The development of the corporate bond market in Korea was more gradual and was initiated by the government with the Capital Market Promotion Act of 1968 that sought to promote both bond and equity financing. This was followed up in the early 1970s by the introduction of guaranteed corporate bonds to ease financial constraints in the face of a major economic downturn. The government ensured that virtually all the corporate bonds issued by the industrial conglomerates (*chaebols*) before the Asian crisis carried bank guarantees to increase their market acceptability, resulting in 85–90 percent of all corporate bonds being guaranteed by financial institutions (Dickie 1999). The corporate bond market grew rapidly in the 1980s and the early 1990s, owing to the relatively large number of reputable industrial conglomerates (chaebols) and the bond guarantee system. By the mid-1990s, the size of bond financing exceeded that from banks and accounted for 28 percent of external financing of corporations (Shiria 2001, Dickie 2000). In the post-Asian crisis period, large corporations have been reducing their reliance on banks by going to the capital markets for their financing, thus freeing up bank credit for smaller firms (Lim 2003).

While the nature and timing of government support differed in Malaysia and Korea, in both cases it was substantial and sustained. The relatively rapid development of bond markets in these two countries seems to suggest that government support is important for bond market development at the initial stage. However, caution must be exercised when designing types of government interventions, as ill-designed government measures may cause many problems.

3. Public Debt Market (+)

The government securities market is an important foundation for corporate bond markets. An active and liquid corporate bond market requires a benchmark yield curve on whose basis risky credit can be priced. The benchmark yield curve is typically constructed from a suite of outstanding treasury securities, requiring governments to issue a range of maturities on a regular schedule. Thus public debt plays an important role in providing the basis for the price discovery of corporate bonds.

This implies that a past history of fiscal deficits funded by the issuance of government securities as well as a liquid secondary market in government securities is a necessary condition for the development of corporate debt markets. Government debt markets also provide dealers with experience in trading fixed income securities, and a chance to earn profit and build credibility as an intermediary.

However, there have been some nascent developments in pricing corporate bond issues off the swap market that have the potential to provide a private benchmark (McCauley 2001). As such, the current dependence on government bond markets for pricing corporate issues may not be permanent. Moreover, large-scale issuance of government bonds could crowd out the issuance of private sector bonds so that the presence of a large public bond sector is not always correlated with a deeper or more active corporate debt market (Gallego and Loayza 2001).

4. Institutional Investors (+)

Financial institutions are often the biggest nongovernment debt purchasers in early stages. Diversified institutional investors composing pension funds, insurance companies, mutual funds, and other financial institutions need to hold long-term debt. They are key to the development of debt markets.

B. Institutions

The importance of institutions for financial development has received major attention in recent years (La Porta et al. 1997).

1. Legal System Origins (?)

Substantial effort has gone into understanding how different legal origins affect financial development. Recent studies highlight the distinction between civil law-based continental European economies and common law-based Anglo-American economies. The French civil law system was intended to promote unified state power with codified legal traditions, while English derived common law systems were intended to protect individuals from the arbitrary actions of the crown and were built on the basis of case law by judges. LaPorta et al. (1998) suggest common law systems in the British tradition are more suited to promoting the development of financial markets because they offer stronger investor protection than the French civil law tradition. La Porta et al. (1997) and Claessens et al. (2000) showed that legal protection of outside investors is associated with larger and more effective stock markets. Beck, Demirguc-Kunt, and Levine (2003) have demonstrated that the English common law has been more adaptable and supportive of financial development relative to the French civil law. Rajan and Zingales (2001) however, argue when the government has the will, civil law countries have a greater ability to translate governmental policy into law because laws emanate from the center rather than evolving through judicial decisions. Private interests have a greater chance of seeing their agenda enacted in a civil law country. Given the mixed result, we do not attempt to postulate, a priori, the relationship between legal system origins and corporate bond market development.

2. Creditor Rights (+)

The protection of minority shareholders' rights is of particular importance for the development of equity markets. Creditor rights and the ability to ensure secured transactions may also be key to the development of debt markets. Shareholders have decision rights over the cash flow when the firm is solvent, but with the onset of financial distress, creditors can acquire those decision rights (Hart 2001). However such transfers of decision rights involve the judiciary in the implementation of defined creditor rights, bringing into play a host of variables such as a country's indices for creditors' rights, regulation of dispute resolution, and judicial efficiency. The influences of most of these variables were explored in La Porta et al. (1997 and 1998).

However, there are many arguments that would lead one to believe that such legal attributes are at least partially endogenous. For example, the development of corporate debt markets may result in enhancing dispute resolution and judicial efficiency through the resolution of commercial disputes (Pistor et al. 2003). In light of this likely endogenity, the legal origin variable was used as an instrument to capture the legal environment that can be thought to be exogenous.

C. Economic Interrelationships and Incentives

Information asymmetries associated with external financing provide important insights into debt financing choices (Myers and Majluf 1984). To avoid information asymmetries, firms tend to rely on internally generated capital. If there is a need for additional external capital, firms tend to rely on debt as opposed to equity, given the large information asymmetries associated with incremental equity issues—the trickle down approach to financing. This suggests that debt markets may be highly opportunistic in responding to spillovers of investment financing requirements in excess of retained earnings. Such spillovers emanate from M&A activity and higher economic growth requiring increased investment (De Bondt 2002; Domowitz, Glen, and Madhaven, 2001).

1. Mergers and Acquisitions (+)

Mergers and acquisitions in one year by foreign investors may invoke issuance of corporate bonds within the following year as a way to control local management's utilization of free cash flow in wasteful investments (Myers 2001). Thus, debt issuance can be concentrated in specific years due to surges in external financial requirements from cross-border M&A activity.

2. Per Capita Income (+)

Economic development and bond market development go hand in hand. The development of corporate debt markets can benefit economic growth through reduced costs of intermediation (Diamond 1994). At the same time, higher incomes can also positively influence bond market development by providing a broader investor base and the investments required to develop supportive institutions such as trading platforms, clearing and settlement systems, regulatory agencies, and credit rating

agencies. Given the likely endogenity of income, a standard geographical measure, namely distance from the equator, is used in alternative regressions (Rodrik, Subramanian, and Trebbi 2002).

3. Defined Contribution Schemes (+)

Recent developments in pension reforms have seen the appearance of defined contribution schemes (DCS), either mandatory or voluntary, to supplement or replace pay-as-you-go (PAYG) systems that are no longer financially viable due to the pressure of aging populations. Bustamente (1996) and Gallego and Loayza (2001) found that Chile's reforms to establish a mandatory DCS in the early 1980s had benefited the development of its equity markets. The presence of DCS creates long-term disbursed savings that may also create demand for corporate debt securities.

4. Corporate Tax Rate (?)

The tradeoff theory states that reliance on debt financing is dependent on the tax deductibility of interest payments in line with Miller's (1977) gains-to-leverage thesis, but that it is also limited by the costs of potential financial distresses as debt levels rise (Myers 2001). MacKie-Mason (1990), utilizing a probit model, found that companies with low marginal tax rates were more willing to issue equity. While the relationship between low tax rates and equity may be unequivocal, it is not obvious that greater issuance of debt securities will be associated with higher corporate tax rates.

IV. DATA AND METHODOLOGY

The limited availability of internationally comparable information on debt securities has been a major constraint in determining the factors associated with debt market development. The data on domestic debt securities from the BIS is only available since 1989 for a gradually increasing number of countries. While the data do not cover all countries issuing debt securities, they do cover the large markets. Data on domestic issuers are segregated into public sector, financial institutions, and corporate issuers, the latter being the focus of this study.

We test the importance of the factors outlined above using annual data for 34 countries/regions² from 1989 to 2002. The dependent variable is corporate bond market capitalization as a share of gross domestic product (GDP) obtained from the BIS database. Data for public debt is also from BIS. Data on equity market capitalization for these years is from the Emerging Stock Markets and the International Finance Corporation. The variable on bank concentration, as measured by the share of the assets of the top three banks to total banking sector assets, is taken from Demirguc-Kunt and Levine (2001). The bank concentration ratios for People's Republic of China (PRC) and Taipei, China

These countries and regions are: Australia; Austria; Belgium; Brazil; Canada; Chile; PRC; Denmark; Finland; France; Germany; Grece; Hong Kong, China; Hungary; Iceland; India; Ireland; Italy; Japan; Korea; Malaysia; Mexico; Netherlands; Norway; New Zealand; Portugal; Singapore; Spain; Sweden; Switzerland; Taipei, China; Turkey; United Kingdom; and United States.

are estimated from national data. Data for GDP per capita, real economic growth, and private sector credit come from *International Financial Statistics* (IFS). Institutional investors are defined as the financial assets of institutional investors as a percentage of GDP and data is extracted from Datastream and the OECD database.

Information on legal and institutional variables is from La Porta et al. (1997) and subsequent papers. Dummy variables were constructed for creditors rights and legal system origins. In addition to the two institutional variables utilized, a large number of alternative variables, including accounting standards, creditor rights, and judicial efficiency were also considered but not included due to their lower explanatory power. The geographical instrument, country latitude, reflects the latitude of the county's capital city, and is normalized between zero and one with the capital farthest from the equator set at one. Corporate tax rates are from Deloitte Australia (personal communication) for the year 2000.

Merger and acquisition data come from the 2003 World Investment Report and reflect cross-border M&A from the economy of the seller as a share of each economy's GDP. The number of global corporations is taken from the Global 500 compilation (Fortune Magazine 1989-2002). In the generalized least squares (GLS) estimation, after some tests, the volume of M&A activity relative to GDP is lagged by one year to reflect the time necessary to put into place debt and equity financing.

Dummy variables are constructed for the defined contribution scheme. Relative to the PAYG scheme, one dummy variable is constructed for mandatory savings schemes in Chile, Denmark, Iceland, Malaysia, and Singapore throughout the period; and the Netherlands and Mexico from 1997 to 2002. Hungary; Hong Kong, China; and Australia comprise the dummy variables for 1998 through 2002. Another dummy variable is set up for countries that have had voluntary defined contribution schemes such as Canada, Ireland, Japan, United Kingdom, and United States for all years. The value for all other countries is assigned as zero (OECD 2000).

Government support refers to countries that used substantial government influence to promote corporate debt markets. Only two countries, Korea and Malaysia are deemed to fall into this group, and thus a dummy variable is used for these two countries. The definitions and sources of the variables are shown in Table 1.

TABLE 1 VARIABLE DEFINITIONS AND SOURCES OF DATA

VARIABLES	DEFINITION	NATURE	SOURCE OF INFORMATION		
CDEBT	Corporate debt outstanding as a percentage of GDP	Dependent variable	BIS		
ACS	Accounting standard	Independent variable	La Porta et al. (1997) and subsequent papers		
JSE	Judicial system efficiency	Independent variable	La Porta et al. (1997) and subsequent papers		
CRD1, CRD2, CRD3*	Creditor rights	Dummy variables	La Porta et al. (1997) and subsequent papers		
LSD1, LSD2, LSD3*	Legal system origin	Dummy variables	La Porta et al. (1997) and subsequent papers		
GOVKM	Administrative support	Dummy variable for Korea and Malaysia			
CTAX	Corporate tax rate	Independent variable	Deloitte		
BCONCT	Bank concentration ratio; share of the assets of the top three banks to total banking sector assets	Independent variable	Demirguc-Kunt and Levine (2001); national sources for PRC and Taipei,China		
LATITUD	Latitude of the county's capital city	Independent varriable			
FORTUN	Number of Global 500 corporations		Fortune magazine		
DCSD1, DCSD2	Mandatory savings schemes, voluntary savings schemes, PAYG, and others	Dummy variables	OECD (2000) and other sources		
GDEBT	Government domestic debt securities as a percentage of GDP	Independent variable	BIS		
MKTCAP	Market capitalization of listed companies (% of GDP)	Independent variable	Emerging Stock Markets, International Finance Corporation		
GDPP	GDP per capita	Independent variable	International Financial Statistics		
GDPG	Real GDP growth rates	Independent variable	International Financial Statistics		
INST	Financial assets of institutional investors as a percentage of GDP	Independent variable	OECD, Datastream		
MA	Cross-border merger and acquisition	Independent variable	World Investment Report		

The analysis uses the panel dataset to take account of the differing interest group incentives and institutional determinants for debt market development. The fixed effects or least squares dummy variables approach is inappropriate in this case as this method requires that there is within cross section variance in all variables for at least some of the cross sections. Equations are estimated using panel GLS with correction for heteroskedasticity and panel-specific autocorrelation.

V. RESULTS

The central hypothesis of this paper is that a highly concentrated banking sector could more effectively protect itself from disintermediation caused by bond market development. The results support this hypothesis in that bank concentration is significantly negatively correlated with bond market development. The finding thus provides support for Rajan and Zingales's (2003) interest group theory of financial development. Banks oppose their own distintermediation that would occur with corporate debt market development. The power of the banks to resist disintermediation is related to their market power. The more concentrated the banking sector, the more negative this association appears to be. The results are presented in Table 2 below.

TABLE 2
REGRESSION RESULTS

VARIABLES	COEFFICIENT	T-VALUE
LSDA	1.679	1.320
GovKM	13.466	5.03**
Ctax	-0.040	-0.559
BCONCT	-6.928	-2.01*
latitud	0.491	0.182
fortun	0.031	1.97*
DCSD1	-0.547	-0.653
DCSD2	1.445	2.14*
GDEBT	0.020	1.480
GDPG	-0.161	-3.39**
INST	0.037	6.51**
MA(-1)	0.207	3.39**
Constant	6.173	2.02*
R-squared	0.39	
Wald (joint): Chi^2(12)	182.8	

^{*} means significant at the 5% level.

Note: The dependent variable is corporate debt securities/GDP (CDEBT); t-ratios in brackets.

^{**} means significant at the 1% level.

The US experience seems to confirm this observation. The US has one of the lowest bank concentration ratios, where the top three banks only constitute about 19 percent of total bank assets, while the ratios of corporate debt market to GDP average a relatively high level of 24 percent of GDP in 2000. Roe (1994) argues that the fragmentation of the financial system in the US as a result of a populist fear of large banks led to government placing restraints on the functional scope and geographical expansion of commercial banks, and to a multifarious financial system involving a large role for capital markets in general and debt markets in particular. By contrast the corresponding bank concentration of the top three banks in the United Kingdom was 89 percent, while its corporate debt market to GDP averaged less than 6 percent in 2000.

The results also confirm that outstanding corporate debt securities as a percentage of GDP are associated with a number of opportunistic elements. The presence of mandatory DCS schemes has a highly significant correctional effect on mandatory schemes. The important influence of mandatory DCS schemes on capital market development was first noted in Chile after their introduction in the early 1980s. The beneficial effects have been subsequently evident in most of countries where such schemes have been introduced to tackle the problem of rising financial burdens on the current generations from the PAYG pension plans in line with aging populations. The availability of long-term savings related to such schemes can contribute to corporate debt market development. For a similar reason, the importance of institutional investors is reflected in the positive and significant correlation with corporate bond market development.

Other elements such as the volume of M&A activity as a share of GDP have also produced the positive signs expected. M&A activity often involves the utilization of high levels of debt in order to control local management that had been prone to overinvest, especially in newly privatized state-owned enterprises. As noted by Myers (2001), this utilization of debt to finance M&A activity provides a solution to Jensen's (1986) free cash flow problem. As the financing is arranged by international purchasers, there is a higher probability that debt markets will be accessed as there may be no relationships with local banks.

The importance of the presence of large global companies as measured by the number of Global Fortune 500 countries is also confirmed by the results. This variable is significant and positive. Large corporations that can issue securities on a "name" or reputation basis provide the key issuers for corporate debt markets. As an optional form of external finance, corporate debt securities issuance responds to unexpected investment requirements that exceed internal cash flows of mainly the larger, creditworthy corporations. In this sense it is also not surprising that the presence of Global 500 corporations within the economy appears to contribute to the development of debt markets, perhaps through overcoming the network effects associated with the development of such markets or in providing a countervailing influence to the banks within the financial sector. Clearly, outward-looking economic policies are supportive of corporate debt market development.

Government debt markets are also positively related to corporate debt market development. Government support for corporate debt market development is captured by the dummy variables for Korea and Malaysia, which shows that government support is associated with corporate debt market development.

Institutional determinants such as accounting standards, creditor rights, and judicial efficiency were used in regressions. However, as noted by Pistor et al. (2003), improvements in the law often come about in response to the challenges posed by the growth of financial sectors. Also, less developed

counties have volatile investment environments and heavy government involvement in commercial activity. Often they have weak creditor rights, inadequate transparency, and poor corporate governance. In light of these problems of endogenity and the correlation between income and the institutional determinants, the legal origin variables are used as an instrumental variable. The English legal origin was found to be positively correlated with corporate debt market development. This is consistent with La Porta et al. (1997 and 1998) who assert that countries with market-based financial systems are much more likely to have English common law origins.

The latitude of the concerned country is utilized as an instrument variable for the exogenous component of income and income-related institutional development. The instrument variable was insignificant. This suggests that income level may not be highly correlated to bond market development.

The sign of the real growth rate of GDP is unexpectedly negative. One reason for this may be that, as investment needs outstrip internal financing during economic expansions, the first recourse for external debt is often to the banks under available credit lines. However, as interest rates fall in the subsequent trough of the economic cycle with the decline in real growth rates, refinancing decisions are undertaken and large-scale bond financing may be implemented at the lower interest rates to reduce bank debt. The effect of corporate tax is also insignificant, suggesting corporate tax rate is not a major factor affecting bond market development.

VI. EQUITY MARKET DEVELOPMENT

To further explore the intrinsic idiosyncrasy of bond market development, the same set of variables is utilized to examine equity market development in the same countries. The intention is mainly to reflect the markedly different associations that these variables have relative to different segments of the financial system. This exercise demonstrates that debt and equity markets are not subject to the same developmental associations.

An important contrast is found with respect to the impact of bank concentration. In the case of equity markets, a higher level of bank concentration ratio is positively related to equity market development. As banks do not generally provide equity financing, there is no conflict of interest in contrast to debt financing that could potentially replace bank credit. Indeed, banks may see equity market listings as a positive development for corporate creditworthiness, given the additional governance oversight provided. The relationship between equity market development and banking sector concentration supports the interest group theory from a different angle.

Another interesting difference is that higher corporate tax rates that were not related to corporate debt market developments are significantly and negatively related to equity market development in line with MacKie-Mason's (1990) findings. The results also show that English legal origin is significant and highly positive, in a result also obtained by La Porta et al. (1997). The volume of M&A activity is significant and positively correlated to equity market development, as for corporate debt market development. However, the importance of the global company is insignificant, suggesting that the presence of large, well-known companies may not be as important a factor as in the case of bond market development. Latitude is negatively related to equity market development, suggesting that countries that are further away from the equator, which tend to be more developed and have higher per capita income, also tend to have more developed equity markets. GDP growth is also positively related to equity market capitalization, as expected.

TABLE 3
REGRESSION RESULTS

VARIABLES	COEFFICIENT	T-VALUE T
LSDA	21.616	1.530
GovKM	-12.005	-0.413
Ctax	-4.730	-6.030**
BCONCT	110.110	*2.890**
latitud	-184.431	-5.990**
fortun	0.151	0.845
DCSD1	-24.664	-2.340*
DCSD2	-2.762	-0.315
GDEBT	0.373	2.210*
GDPG	2.736	4.520**
INST	0.870	12.300**
MA(-1)	1.954	2.480*
Constant	181.703	5.290
R-squared	0.572	
Wald (joint): Chi^2(12)	362.6	

^{*} means variables are significant at the 5% level

Note: The dependent is equity market capitalization/GDP (MKTCAT); t-ratios in brackets.

VII. CONCLUSIONS AND POLICY IMPLICATIONS

This study is an initial attempt to reflect factors that are associated with the development of corporate debt markets. It particularly seeks to explore the relationship between banking sector structure and corporate bond market development. The results support Rajan and Zingales' (2003) interest group theory of financial development. The study provides some evidence that banks oppose corporate debt market development because of the distintermediation they would suffer as a result. Banks are likely to resist the development of debt markets as they have the potential of providing competition across a full range of debt maturities including, for example, long-term corporate bond debt as well as short-term commercial paper debt. Thus, there is competition between banks and bond markets in providing external finance. Banks are a key interest group in the development of corporate debt markets due to the effects the latter have on their disintermediation, as corporate bonds can be viewed as a loan substitute. In some circumstances, highly concentrated banking systems may succeed in depriving bonds of market development.

^{**} means variables are significant at the 1% level.

Corporate debt markets can only develop in an environment where the natural resistance of banks can be overcome. Substantial progress in developing local currency corporate debt markets may not be possible without addressing issues associated with the position of incumbent banks in the financial sector. Banks have often been supported by central banks or other government agencies that have the power to approve bond issues. Rajan and Zingales (2003) note that in an earlier period Japanese banks withheld the access of corporations to the bond market in Japan through their control of the Bond Committee. The Committee was formalized in 1933 and determined which companies could issue bonds, on what terms, and when. Partly due to the influence of the committee, the Japanese bond market shrank. In 1929, the liability side of large Japanese company balance sheets consisted of 26 percent bonds and 17 percent of bank debt. By 1943, the percentage of bonds had fallen to 6 percent, while that of bank debt rose to 47 percent.

Several factors can countervail the power of banks. Sustained governmental support can result in major gains for the development of such markets. This is not only historically evident in the US, but also in the rapid development of corporate debt markets in Korea and Malaysia that occurred during the 1990s. While the nature of the support varied between the two countries, both governments provided support to ensure that an adequate level of infrastructure was present for the debt markets, intervened to support the funding of specific issues, and leaned on the banks to guarantee most of the bond issues. With these governmental programs in place, the corporate debt market in Malaysia rose from 4.4 percent of GDP in 1989 to 47.7 percent in 2000; while in Korea, the corporate debt market rose from 11.1 percent of GDP in 1989 to 26.0 percent in 2000.

The Korean and Malaysian experience shows that governmental support can substantially overcome the influence of commercial banks. However, there are also risks to the integrity of the markets from such government intervention. For example, the widespread utilization of banks to guarantee corporate bonds in Korea and Malaysia may have had a negative impact on the creditworthiness of those banks during the Asian crisis. The relative importance and risks of governmental actions to developing corporate debt markets is a subject warranting further research.

The active involvement of banks in the development of debt markets is perhaps a good way to overcome their opposition to bond market development without the risks of administrative directions from the governments concerned. One of the potential approaches to enlist commercial bank support for corporate debt market development is to encourage them to lead the effort in niche markets. For example, mortgage-backed securities allow banks to fully participate in debt market development through mortgage origination fees. Further, securitization of mortgages takes these longer-term loans off banks' balance sheets and reduces their maturity mismatches. Such approaches may lessen bank opposition to the development of debt markets. This secondary mortgage market approach to soliciting the cooperation of banks originated in Malaysia and is now being implemented in Hong Kong, China. At the shorter end of the market, banks have usefully participated in the securitization of receivables, such as those that originate from their credit card issuance.

The development in securitization of mortgage and consumer credits blurs the traditional distinction between intermediation through a bank, which typically acquires long-term nonmarketable loans held on the balance sheets until maturity, and that through capital markets where assets trade in secondary markets (Turner 2002). These developments suggest that the bank-bond market development relationship

needs to be viewed in a dynamic sense. While banks may oppose the development of bond markets initially, this relationship may evolve and change over time.

In the long run, banks can survive only if they adapt and learn to play a major role in capital markets. Over time, banks can evolve as issuers, investors, and intermediaries of bond markets. For example, in Indonesia banks hold the majority of corporate bonds. This used to be the case in Argentina, Brazil, Chile, and Malaysia as well. Some banks derive more profit from such activities and less from lending. For this reason, it is important to have healthy banks as potential contributors to a sound bond market, while a bond market may improve the health of banks by improving market discipline (Hawkins 2002). This suggests that a complementary relationship between banks and corporate debt market development may come into being.

Outward-looking economic policies will also help bring pressures to develop debt markets in order to reduce capital costs and associated risks. This is highlighted though the impact of cross-border M&A activity on their development. Corporate debt securities issuance responds to unexpected investment requirements that exceed internal cash flows of larger, creditworthy corporations. In this sense it is also not surprising that the presence of Global 500 corporations within the economy appears to contribute to the development of debt markets, perhaps through overcoming the network effects associated with the development of such markets or in providing a countervailing influence to the banks within the financial sector.

Defined contribution saving schemes also benefit corporate debt market development, most likely due to the long-term and distributed nature of the savings that are generated. This is a beneficial spillover from solving the deficiencies of the pay-as-you-go pension systems under pressures of aging populations. Not surprisingly, the development of institutional investors also benefits bond market development.

It is also important to note that different factors have different impacts on corporate debt markets and equity markets respectively, so that caution needs to be exercised in making assumptions about different components of the financial system.

REFERENCES

- Beck, T., A. Demirguc-Kunt, and R. Levine, 2003. "Law and Finance: Why Does Legal Origin Matter?" *Journal of Comparative Economics* 31(December, 4):653-75.
- Bolton, P., and X. Freixas, 2000. "Equity, Bonds, and Bank Debt: Capital Structure and Financial Market Equilibrium Under Asymmetric Information." *Journal of Political Economy* 108:324-51.
- Bustamente, J.,1996. *The Chilean Pension System*. Superintendence of Pension Funds Administrators, Santiago, Chile.
- Claessens, S., S. Djankov, J. Fan, and L. Lang, 2000. Expropriation of Minority Shareholders in East Asia. Working Paper 2088, World Bank, Washington, DC.
- Dalla, I., 1995. The Emerging Asian Bond Market. World Bank, Washington, DC.
- De Bondt, G., 2002. Euro Area Corporate Debt Securities Market: First Empirical Evidence. Working Paper No. 164, European Central Bank, Frankfurt, Germany.
- Demirguc-Kunt, A., and H. Huizinga, 2001. Financial Structure and Bank Profitability, in Financial Structure and Economic Growth: A Cross-country Comparison of Banks, Markets, and Development. Cambridge and London: MIT Press.
- Demirguc-Kunt, A., and R. Levine, 2001. Financial Structure and Economic Growth. Cambridge, MA: MIT Press.

- Denis, D. J., and V. T. Mihov, 2003. "The Choice among Bank Debt, Non-bank Private Debt and Public Debt: Evidence from New Corporate Borrowings." *Journal of Financial Economics* 70(October, 1)3-28.
- Diamond, D., 1994. "Corporate Capital Structure: The Control Roles of Bank and Public Debt with Taxes and Costly Bankruptcy." Federal Reserve Bank of Richmond Economic Quarterly 80:11-37.
- Dickie, P., 1999. Strengthening East Asian Financial Systems: An Integrated Approach. Asian Studies Institute Working Paper 10, Victoria University of Wellington, New Zealand.
- ______, 2000. Corporate Debt/Equity Ratios and the Korean Financial Crisis. Graduate School of Business and Government Management Working Paper 3/00, Victoria University of Wellington, New Zealand.
- Domowitz, I., J. Glen, and A. Madhaven, 2001. "International Evidence on Aggregate Corporate Financing Decisions." In A. Demirguc-Kunt and R. Levine, eds., *Financial Structure and Economic Growth*. Cambridge, MA: MIT Press.
- Dornbusch, R., 2002. "The New International Architecture." FinanzArchiv 59(3):296-304.
- Eichengreen, B., and R. Haussmann, 1999. "Exchange Rates and Financial Fragility." *New Challenges for Monetary Policy* 329-68.
- Eichengreen, B., and P. Luengnaruemitchai, 2004. Why Doesn't Asia Have Bigger Bond Markets? NBER Working Paper No. w10576, National Bureau of Economic Research, Massachusetts.
- Fergusion, N., 2001. The Cash Nexus. New York: Basic Books.
- Gallego, F., and N. Loayza, 2001. "Financial Structure in Chile: Macroeconomic Developments and Microeconomic Effects." In A. Demirguc-Kunt and R. Levine, eds., *Financial Structure and Economic Growth*. Cambridge, MA: MIT Press.
- Gandal, N., 2002. "Compatibility, Standardization, and Network Effects: Some Policy Implications." Oxford Review of Economic Policy 18(1):80-91.
- Hart, O., 2001. "Financial Contracting." Journal of Economic Literature 39:1079-100.
- Hausmann, R., U. Panizza, and E. Stein, 2001. "Why Do Countries Float the Way they Float?" *Journal of Development Economics* 66(December):387-414.
- Hawkins, J., 2002. "Bond Markets and Banks in Emerging Economies." In The Development of Bond Markets in Emerging Economies, BIS Papers No. 11, Monetary and Economic Department, Bank for International Settlements.
- Jensen, M. C., 1986. "Agency Costs of Free Cash Flow, Corporate Finance and Takeovers." *American Economic Review* 76:323-39.
- Jiang, G., N. Tang, and E. Law, 2001, "Cost and Benefit Analysis of Developing Debt Markets." Hong Kong Monetary Authority Quarterly Bulletin 29(November): 1-18. Available: www.infor.gov.hk/hkma.
- Knight, M., 1998. Developing Countries and the Globalization of Financial Markets. IMF Working Paper WP/98/105, International Monetary Fund, Washington, DC.
- LaPorta, R., F. Lopez-de-Silanes, A. Shleifer, and R.Vishny, 1997. "Legal Determinants of External Finance." *The Journal of Finance* 52(3):1131-49.
- ______, 1998. "Law and Finance." Journal of Political Economy 106:1113-55.
- Lim, Y., 2003. Sources of Corporate Financing and Economic Crisis in Korea: A Micro-Evidence. NBER Working Paper 9575, National Bureau of Economic Research, Massachusetts.
- Mackie-Mason, J. K., 1990. "Do Taxes Affect Corporate Financing Decisions?" *Journal of Finance* 45:1471-93. Mc Cauley, R. N., 2001. "Benchmark Tipping in the Money and Bond Markets." *BIS Quarterly Review* March:39-45.
- Miller, M., 1977. "Debt and Taxes." Journal of Finance 32:261-75.
- Myers, S., 2001. "Capital Structure." Journal of Economic Perspectives 15(2):81-102.

PUBLICATIONS FROM THE ECONOMICS AND RESEARCH DEPARTMENT

ERD WORKING PAPER SERIES (WPS) (Published in-house; Available through ADB Office of External Relations; Free of Charge)

No. 1	Capitalizing on Globalization —Barry Eichengreen, January 2002	No. 20	Modernization and Son Preference in People's Republic of China
No. 2	Policy-based Lending and Poverty Reduction:		—Robin Burgess and Juzhong Zhuang, September
	An Overview of Processes, Assessment	No. 91	The Debe Agenda and Development, A View from
	and Options —Richard Bolt and Manabu Fujimura, January	No. 21	The Doha Agenda and Development: A View from the Uruguay Round
	2002		—J. Michael Finger, September 2002
No. 3	The Automotive Supply Chain: Global Trends	No. 22	Conceptual Issues in the Role of Education
110. 5	and Asian Perspectives	110. 22	Decentralization in Promoting Effective Schooling in
	—Francisco Veloso and Rajiv Kumar, January 2002		Asian Developing Countries
No. 4	International Competitiveness of Asian Firms:		—Jere R. Behrman, Anil B. Deolalikar, and Lee-
110. 4	An Analytical Framework		Ying Son, September 2002
	—Rajiv Kumar and Doren Chadee, February 2002	No. 23	Promoting Effective Schooling through Education
No. 5	The International Competitiveness of Asian	110. 20	Decentralization in Bangladesh, Indonesia, and
110. 0	Economies in the Apparel Commodity Chain		Philippines
	—Gary Gereffi, February 2002		—Jere R. Behrman, Anil B. Deolalikar, and Lee-
No. 6	Monetary and Financial Cooperation in East		Ying Son, September 2002
	Asia—The Chiang Mai Initiative and Beyond	No. 24	Financial Opening under the WTO Agreement in
	—Pradumna B. Rana, February 2002		Selected Asian Countries: Progress and Issues
No. 7	Probing Beneath Cross-national Averages: Poverty,		—Yun-Hwan Kim, September 2002
	Inequality, and Growth in the Philippines	No. 25	Revisiting Growth and Poverty Reduction in
	—Arsenio M. Balisacan and Ernesto M. Pernia,		Indonesia: What Do Subnational Data Show?
	March 2002		—Arsenio M. Balisacan, Ernesto M. Pernia,
No. 8	Poverty, Growth, and Inequality in Thailand		and Abuzar Asra, October 2002
	—Anil B. Deolalikar, April 2002	No. 26	Causes of the 1997 Asian Financial Crisis: What
No. 9	Microfinance in Northeast Thailand: Who Benefits		Can an Early Warning System Model Tell Us?
	and How Much?		—Juzhong Zhuang and J. Malcolm Dowling,
	—Brett E. Coleman, April 2002		October 2002
No. 10	Poverty Reduction and the Role of Institutions in	No. 27	Digital Divide: Determinants and Policies with
	Developing Asia		Special Reference to Asia
	—Anil B. Deolalikar, Alex B. Brilliantes, Jr.,		—M. G. Quibria, Shamsun N. Ahmed, Ted
	Raghav Gaiha, Ernesto M. Pernia, Mary Racelis		Tschang, and Mari-Len Reyes-Macasaquit, October 2002
	with the assistance of Marita Concepcion Castro- Guevara, Liza L. Lim, Pilipinas F. Quising, May	No. 28	Regional Cooperation in Asia: Long-term Progress,
	2002	110. 20	Recent Retrogression, and the Way Forward
No. 11	The European Social Model: Lessons for		—Ramgopal Agarwala and Brahm Prakash,
110. 11	Developing Countries		October 2002
	—Assar Lindbeck, May 2002	No. 29	How can Cambodia, Lao PDR, Myanmar, and Viet
No. 12	Costs and Benefits of a Common Currency for	110. 20	Nam Cope with Revenue Lost Due to AFTA Tariff
	ASEAN		Reductions?
	—Srinivasa Madhur, May 2002		-Kanokpan Lao-Araya, November 2002
No. 13	Monetary Cooperation in East Asia: A Survey	No. 30	Asian Regionalism and Its Effects on Trade in the
	—Raul Fabella, May 2002		1980s and 1990s
No. 14	Toward A Political Economy Approach		-Ramon Clarete, Christopher Edmonds, and
	to Policy-based Lending		Jessica Seddon Wallack, November 2002
	—George Abonyi, May 2002	No. 31	New Economy and the Effects of Industrial
No. 15	A Framework for Establishing Priorities in a		Structures on International Equity Market
	Country Poverty Reduction Strategy		Correlations
	—Ron Duncan and Steve Pollard, June 2002		—Cyn-Young Park and Jaejoon Woo, December
No. 16	The Role of Infrastructure in Land-use Dynamics		2002
	and Rice Production in Viet Nam's Mekong River	No. 32	Leading Indicators of Business Cycles in Malaysia
	Delta		and the Philippines
N. 15	—Christopher Edmonds, July 2002		—Wenda Zhang and Juzhong Zhuang, December
No. 17	Effect of Decentralization Strategy on	NT 00	2002
	Macroeconomic Stability in Thailand	No. 33	Technological Spillovers from Foreign Direct
No. 10	—Kanokpan Lao-Araya, August 2002		Investment—A Survey
No. 18	Poverty and Patterns of Growth —Rana Hasan and M. G. Quibria, August 2002	No. 34	—Emma Xiaoqin Fan, December 2002
No. 19	Why are Some Countries Richer than Others?	110. 54	Economic Openness and Regional Development in the Philippines
110. 19	A Reassessment of Mankiw-Romer-Weil's Test of		—Ernesto M. Pernia and Pilipinas F. Quising,
	the Neoclassical Growth Model		January 2003
	—Jesus Felipe and John McCombie, August 2002	No. 35	Bond Market Development in East Asia:
	3 como I cupe ana 3 onn miconnoue, magasi 2002	110. 00	Zona Markov Development in Dast Itsia.

- Issues and Challenges
 —Raul Fabella and Srinivasa Madhur, January
 2003
- No. 36 Environment Statistics in Central Asia: Progress and Prospects

 —Robert Ballance and Bishnu D. Pant, March
- No. 37 Electricity Demand in the People's Republic of China: Investment Requirement and Environmental Impact —Bo Q. Lin, March 2003
- No. 38 Foreign Direct Investment in Developing Asia: Trends, Effects, and Likely Issues for the Forthcoming WTO Negotiations —Douglas H. Brooks, Emma Xiaoqin Fan, and Lea R. Sumulong, April 2003
- No. 39 The Political Economy of Good Governance for Poverty Alleviation Policies —Narayan Lakshman, April 2003
- No. 40 The Puzzle of Social Capital A Critical Review
 —M. G. Quibria, May 2003
- No. 41 Industrial Structure, Technical Change, and the Role of Government in Development of the Electronics and Information Industry in Taipei, China

 —Yeo Lin, May 2003
- No. 42 Economic Growth and Poverty Reduction in Viet Nam
 —Arsenio M. Balisacan, Ernesto M. Pernia, and Gemma Esther B. Estrada, June 2003
- No. 43 Why Has Income Inequality in Thailand Increased? An Analysis Using 1975-1998 Surveys —Taizo Motonishi, June 2003
- No. 44 Welfare Impacts of Electricity Generation Sector Reform in the Philippines —Natsuko Toba, June 2003
- No. 45 A Review of Commitment Savings Products in Developing Countries —Nava Ashraf, Nathalie Gons, Dean S. Karlan, and Wesley Yin, July 2003
- No. 46 Local Government Finance, Private Resources, and Local Credit Markets in Asia —Roberto de Vera and Yun-Hwan Kim, October 2003
- No. 47 Excess Investment and Efficiency Loss During Reforms: The Case of Provincial-level Fixed-Asset Investment in People's Republic of China —Duo Qin and Haiyan Song, October 2003
- No. 48 Is Export-led Growth Passe? Implications for Developing Asia

 —Jesus Felipe, December 2003
- No. 49 Changing Bank Lending Behavior and Corporate Financing in Asia—Some Research Issues —Emma Xiaoqin Fan and Akiko Terada-Hagiwara, December 2003
- No. 50 Is People's Republic of China's Rising Services Sector Leading to Cost Disease? —Duo Qin, March 2004
- No. 51 Poverty Estimates in India: Some Key Issues
 —Savita Sharma, May 2004
- No. 52 Restructuring and Regulatory Reform in the Power Sector: Review of Experience and Issues —Peter Choynowski, May 2004
- No. 53 Competitiveness, Income Distribution, and Growth in the Philippines: What Does the Long-run Evidence Show?

 —Jesus Felipe and Grace C. Sipin, June 2004
- No. 54 Practices of Poverty Measurement and Poverty Profile of Bangladesh —Faizuddin Ahmed, August 2004
- No. 55 Experience of Asian Asset Management Companies: Do They Increase Moral Hazard? —Evidence from Thailand

- —Akiko Terada-Hagiwara and Gloria Pasadilla, September 2004
- No. 56 Viet Nam: Foreign Direct Investment and Postcrisis Regional Integration —Vittorio Leproux and Douglas H. Brooks, September 2004
- No. 57 Practices of Poverty Measurement and Poverty Profile of Nepal —Devendra Chhetry, September 2004
- No. 58 Monetary Poverty Estimates in Sri Lanka: Selected Issues
 —Neranjana Gunetilleke and Dinushka Senanayake, October 2004
- No. 59 Labor Market Distortions, Rural-Urban Inequality, and the Opening of People's Republic of China's Economy
- —Thomas Hertel and Fan Zhai, November 2004
 No. 60 Measuring Competitiveness in the World's Smallest Economies: Introducing the SSMECI
 —Ganeshan Wignaraja and David Joiner, November 2004
- No. 61 Foreign Exchange Reserves, Exchange Rate Regimes, and Monetary Policy: Issues in Asia —Akiko Terada-Hagiwara, January 2005
- No. 62 A Small Macroeconometric Model of the Philippine Economy
 —Geoffrey Ducanes, Marie Anne Cagas, Duo Qin, Pilipinas Quising, and Nedelyn Magtibay-Ramos, January 2005
- No. 63 Developing the Market for Local Currency Bonds by Foreign Issuers: Lessons from Asia —Tobias Hoschka, February 2005
- No. 64 Empirical Assessment of Sustainability and Feasibility of Government Debt: The Philippines Case

 —Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Nedelyn Magtibay-Ramos, and Pilipinas Quising, February 2005
- No. 65 Poverty and Foreign Aid
 Evidence from Cross-Country Data
 —Abuzar Asra, Gemma Estrada, Yangseom Kim,
 and M. G. Quibria, March 2005
- No. 66 Measuring Efficiency of Macro Systems: An Application to Millennium Development Goal Attainment
- —Ajay Tandon, March 2005
 No. 67 Banks and Corporate Debt Market Development
 —Paul Dickie and Emma Xiaoqin Fan, April 2005

ERD TECHNICAL NOTE SERIES (TNS)

(Published in-house; Available through ADB Office of External Relations; Free of Charge)

- $\begin{array}{ccc} \text{No. 1} & \text{Contingency Calculations for Environmental} \\ & \text{Impacts with Unknown Monetary Values} \\ & -David \ Dole, \ February \ 2002 \\ \text{No. 2} & \text{Integrating Risk into ADB's Economic Analysis} \end{array}$
- No. 2 Integrating Risk into ADB's Economic Analysis of Projects

 —Nigel Rayner, Anneli Lagman-Martin, and Keith Ward, June 2002
- No. 3 Measuring Willingness to Pay for Electricity —Peter Choynowski, July 2002
- No. 4 Economic Issues in the Design and Analysis of a Wastewater Treatment Project
 —David Dole, July 2002
- No. 5 An Analysis and Case Study of the Role of Environmental Economics at the Asian Development Bank —David Dole and Piya Abeygunawardena, September 2002
- No. 6 Economic Analysis of Health Projects: A Case Study in Cambodia
 - —Erik Bloom and Peter Choynowski, May 2003

- No. 7 Strengthening the Economic Analysis of Natural Resource Management Projects —Keith Ward, September 2003
- No. 8 Testing Savings Product Innovations Using an Experimental Methodology

 —Nava Ashraf, Dean S. Karlan, and Wesley Yin, November 2003
- No. 9 Setting User Charges for Public Services: Policies and Practice at the Asian Development Bank
 —David Dole, December 2003
- No. 10 Beyond Cost Recovery: Setting User Charges for Financial, Economic, and Social Goals —David Dole and Ian Bartlett, January 2004
- No. 11 Shadow Exchange Rates for Project Economic Analysis: Toward Improving Practice at the Asian Development Bank —Anneli Lagman-Martin, February 2004

ERD POLICY BRIEF SERIES (PBS)

(Published in-house; Available through ADB Office of External Relations; Free of charge)

- No. 1 Is Growth Good Enough for the Poor?

 —Ernesto M. Pernia, October 2001
- No. 2 India's Economic Reforms
 What Has Been Accomplished?
 What Remains to Be Done?
 —Arvind Panagariya, November 2001
- No. 3 Unequal Benefits of Growth in Viet Nam

 —Indu Bhushan, Erik Bloom, and Nguyen Minh
 Thang, January 2002
- No. 4 Is Volatility Built into Today's World Economy?

 —J. Malcolm Dowling and J.P. Verbiest,
 February 2002
- No. 5 What Else Besides Growth Matters to Poverty Reduction? Philippines —Arsenio M. Balisacan and Ernesto M. Pernia, February 2002
- No. 6 Achieving the Twin Objectives of Efficiency and Equity: Contracting Health Services in Cambodia
 —Indu Bhushan, Sheryl Keller, and Brad Schwartz,
 March 2002
- No. 7 Causes of the 1997 Asian Financial Crisis: What Can an Early Warning System Model Tell Us?

 —Juzhong Zhuang and Malcolm Dowling,

 June 2002
- No. 8 The Role of Preferential Trading Arrangements in Asia

 —Christopher Edmonds and Jean-Pierre Verbiest,
 July 2002
- No. 9 The Doha Round: A Development Perspective
 —Jean-Pierre Verbiest, Jeffrey Liang, and Lea
 Sumulong, July 2002
- No. 10 Is Economic Openness Good for Regional Development and Poverty Reduction? The Philippines

 —E. M. Pernia and Pilipinas Quising, October 2002

- No. 11 Implications of a US Dollar Depreciation for Asian Developing Countries —Emma Fan, July 2002
- No. 12 Dangers of Deflation

 —D. Brooks and Pilipinas Quising, December 2002
- No. 13 Infrastructure and Poverty Reduction— What is the Connection? —Ifzal Ali and Ernesto Pernia, January 2003
- No. 14 Infrastructure and Poverty Reduction—
 Making Markets Work for the Poor
 —Xianbin Yao, May 2003
- No. 15 SARS: Economic Impacts and Implications
 —Emma Xiaoqin Fan, May 2003
- No. 16 Emerging Tax Issues: Implications of Globalization and Technology

 —Kanokpan Lao Araya, May 2003
- No. 17 Pro-Poor Growth: What is It and Why is It Important?
 —Ernesto M. Pernia, May 2003
- No. 18 Public-Private Partnership for Competitiveness
 Jesus Felipe, June 2003
- No. 19 Reviving Asian Economic Growth Requires Further Reforms
 —Ifzal Ali, June 2003
- No. 20 The Millennium Development Goals and Poverty: Are We Counting the World's Poor Right? —M. G. Quibria, July 2003
- No. 21 Trade and Poverty: What are the Connections?

 —Douglas H. Brooks, July 2003
- No. 22 Adapting Education to the Global Economy
 —Olivier Dupriez, September 2003
- No. 23 Avian Flu: An Economic Assessment for Selected Developing Countries in Asia —Jean-Pierre Verbiest and Charissa Castillo, March 2004

- No. 25 Purchasing Power Parities and the International Comparison Program in a Globalized World
 —Bishnu Pant, March 2004

 No. 26 A Note on Dual/Multiple Exchange Rates
 —Emma Xiaoqin Fan, May 2004
- No. 27 Inclusive Growth for Sustainable Poverty Reduction in Developing Asia: The Enabling Role of Infrastructure Development

 —Ifzal Ali and Xianbin Yao, May 2004
- No. 28 Higher Oil Prices: Asian Perspectives and Implications for 2004-2005

 —Cyn-Young Park, June 2004
- No. 29 Accelerating Agriculture and Rural Development for Inclusive Growth: Policy Implications for Developing Asia

 —Richard Bolt, July 2004
- No. 30 Living with Higher Interest Rates: Is Asia Ready?

 —Cyn-Young Park, August 2004
- No. 31 Reserve Accumulation, Sterilization, and Policy Dilemma
 - —Akiko Terada-Hagiwara, October 2004

- 32 The Primacy of Reforms in the Emergence of People's Republic of China and India
 —Ifzal Ali and Emma Xiaoqin Fan, November 2004
- Population Health and Foreign Direct Investment:
 Does Poor Health Signal Poor Government
 Effectiveness?
 —Ajay Tandon, January 2005
- 34 Financing Infrastructure Development: Asian Developing Countries Need to Tap Bond Markets More Rigorously

 —Yun-Hwan Kim, February 2005
- 35 Attaining Millennium Development Goals in Health: Isn't Economic Growth Enough?

 —Ajay Tandon, March 2005
- 36 Instilling Credit Culture in State-owned Banks— Experience from Lao PDR—Robert Boumphrey, Paul Dickie, and Samiuela Tukuafu, March 2005

SPECIAL STUDIES, COMPLIMENTARY

(Available through ADB Office of External Relations)

- 1. Improving Domestic Resource Mobilization Through Financial Development: Overview September 1985
- 2. Improving Domestic Resource Mobilization Through Financial Development: Bangladesh $July\ 1986$
- 3. Improving Domestic Resource Mobilization Through Financial Development: Sri Lanka April 1987
- Improving Domestic Resource Mobilization Through Financial Development: India December 1987
- Financing Public Sector Development Expenditure in Selected Countries: Overview January 1988
- 6. Study of Selected Industries: A Brief Report
 April 1988
- Financing Public Sector Development Expenditure in Selected Countries: Bangladesh June 1988
- 8. Financing Public Sector Development Expenditure
- in Selected Countries: India June 1988
 9. Financing Public Sector Development Expenditure in Selected Countries: Indonesia June 1988
- Financing Public Sector Development Expenditure in Selected Countries: Nepal June 1988
- 11. Financing Public Sector Development Expenditure in Selected Countries: Pakistan June 1988
- 12. Financing Public Sector Development Expenditure
- in Selected Countries: Philippines *June 1988*13. Financing Public Sector Development Expenditure
- in Selected Countries: Thailand *June 1988*14. Towards Regional Cooperation in South Asia:
- ADB/EWC Symposium on Regional Cooperation in South Asia:

 ADB/EWC Symposium on Regional Cooperation in South Asia February 1988
- 15. Evaluating Rice Market Intervention Policies: Some Asian Examples $April\ 1988$
- 16. Improving Domestic Resource Mobilization Through Financial Development: Nepal November 1988
- 17. Foreign Trade Barriers and Export Growth September 1988

- The Role of Small and Medium-Scale Industries in the Industrial Development of the Philippines April 1989
- The Role of Small and Medium-Scale Manufacturing Industries in Industrial Development: The Experience of Selected Asian Countries January 1990
- National Accounts of Vanuatu, 1983-1987 January 1990
- National Accounts of Western Samoa, 1984-1986 February 1990
- 22. Human Resource Policy and Economic Development: Selected Country Studies *July 1990*
- 23. Export Finance: Some Asian Examples September 1990
- National Accounts of the Cook Islands, 1982-1986
 September 1990
- Framework for the Economic and Financial Appraisal of Urban Development Sector Projects January 1994
- Framework and Criteria for the Appraisal and Socioeconomic Justification of Education Projects January 1994
- 27. Investing in Asia 1997 (Co-published with OECD)
- The Future of Asia in the World Economy 1998 (Copublished with OECD)
- 29. Financial Liberalisation in Asia: Analysis and Prospects 1999~(Co-published~with~OECD)
- 30. Sustainable Recovery in Asia: Mobilizing Resources for Development 2000 (Co-published with OECD)
- 31. Technology and Poverty Reduction in Asia and the Pacific 2001 (Co-published with OECD)
- 32. Asia and Europe 2002 (Co-published with OECD)
- 33. Economic Analysis: Retrospective 2003
- 34. Economic Analysis: Retrospective: 2003 Update 2004
- Development Indicators Reference Manual: Concepts and Definitions 2004

OLD MONOGRAPH SERIES

(Available through ADB Office of External Relations; Free of charge)

EDRC REPORT SERIES (ER)

No. 1	ASEAN and the Asian Development Bank	No. 22	Effects of External Shocks on the Balance
No. 2	—Seiji Naya, April 1982 Development Issues for the Developing East		of Payments, Policy Responses, and Debt Problems of Asian Developing Countries
	and Southeast Asian Countries		—Seiji Naya, December 1983
	and International Cooperation	No. 23	Changing Trade Patterns and Policy Issues:
No. 3	—Seiji Naya and Graham Abbott, April 1982 Aid, Savings, and Growth in the Asian Region		The Prospects for East and Southeast Asian Developing Countries
110. 5	—J. Malcolm Dowling and Ulrich Hiemenz,		—Seiji Naya and Ulrich Hiemenz, February 1984
	April 1982	No. 24	Small-Scale Industries in Asian Economic
No. 4	Development-oriented Foreign Investment		Development: Problems and Prospects
	and the Role of ADB		—Seiji Naya, February 1984
	—Kiyoshi Kojima, April 1982	No. 25	A Study on the External Debt Indicators
No. 5	The Multilateral Development Banks		Applying Logit Analysis
	and the International Economy's Missing Public Sector	No. 26	—Jungsoo Lee and Clarita Barretto, February 1984 Alternatives to Institutional Credit Programs
	—John Lewis, June 1982	110. 20	in the Agricultural Sector of Low-Income
No. 6	Notes on External Debt of DMCs		Countries
	—Evelyn Go, July 1982		—Jennifer Sour, March 1984
No. 7	Grant Element in Bank Loans	No. 27	Economic Scene in Asia and Its Special Features
	—Dal Hyun Kim, July 1982		-Kedar N. Kohli, November 1984
No. 8	Shadow Exchange Rates and Standard	No. 28	The Effect of Terms of Trade Changes on the
	Conversion Factors in Project Evaluation		Balance of Payments and Real National
No. 9	—Peter Warr, September 1982 Small and Medium-Scale Manufacturing		Income of Asian Developing Countries —Jungsoo Lee and Lutgarda Labios, January 1985
110. 0	Establishments in ASEAN Countries:	No. 29	Cause and Effect in the World Sugar Market:
	Perspectives and Policy Issues		Some Empirical Findings 1951-1982
	—Mathias Bruch and Ulrich Hiemenz, January		—Yoshihiro Iwasaki, February 1985
3. 4.0	1983	No. 30	Sources of Balance of Payments Problem
No. 10	A Note on the Third Ministerial Meeting of GATT		in the 1970s: The Asian Experience
No. 11	—Jungsoo Lee, January 1983 Macroeconomic Forecasts for the Republic	No. 31	—Pradumna Rana, February 1985 India's Manufactured Exports: An Analysis
110. 11	of China, Hong Kong, and Republic of Korea	110. 51	of Supply Sectors
	—J.M. Dowling, January 1983		—Ifzal Ali, February 1985
No. 12	ASEAN: Economic Situation and Prospects	No. 32	Meeting Basic Human Needs in Asian
	—Seiji Naya, March 1983		Developing Countries
No. 13	The Future Prospects for the Developing	N. 00	—Jungsoo Lee and Emma Banaria, March 1985
	Countries of Asia	No. 33	The Impact of Foreign Capital Inflow
No. 14	—Seiji Naya, March 1983 Energy and Structural Change in the Asia-		on Investment and Economic Growth in Developing Asia
110. 11	Pacific Region, Summary of the Thirteenth		—Evelyn Go, May 1985
	Pacific Trade and Development Conference	No. 34	The Climate for Energy Development
	—Seiji Naya, March 1983		in the Pacific and Asian Region:
No. 15	A Survey of Empirical Studies on Demand		Priorities and Perspectives
	for Electricity with Special Emphasis on Price	NI 05	—V.V. Desai, April 1986
	Elasticity of Demand —Wisarn Pupphavesa, June 1983	No. 35	Impact of Appreciation of the Yen on Developing Member Countries of the Bank
No. 16	Determinants of Paddy Production in Indonesia:		—Jungsoo Lee, Pradumna Rana, and Ifzal Ali,
110. 10	1972-1981–A Simultaneous Equation Model		May 1986
	Approach	No. 36	Smuggling and Domestic Economic Policies
	—T.K. Jayaraman, June 1983		in Developing Countries
No. 17	The Philippine Economy: Economic	N. 05	—A.H.M.N. Chowdhury, October 1986
	Forecasts for 1983 and 1984	No. 37	Public Investment Criteria: Economic Internal
	—J.M. Dowling, E. Go, and C.N. Castillo, June 1983		Rate of Return and Equalizing Discount Rate —Ifzal Ali, November 1986
No. 18	Economic Forecast for Indonesia	No. 38	Review of the Theory of Neoclassical Political
	—J.M. Dowling, H.Y. Kim, Y.K. Wang,		Economy: An Application to Trade Policies
	and C.N. Castillo, June 1983		—M.G. Quibria, December 1986
No. 19	Relative External Debt Situation of Asian	No. 39	Factors Influencing the Choice of Location:
	Developing Countries: An Application		Local and Foreign Firms in the Philippines
	of Ranking Method	No. 40	—E.M. Pernia and A.N. Herrin, February 1987
No. 20	—Jungsoo Lee, June 1983 New Evidence on Yields, Fertilizer Application,	No. 40	A Demographic Perspective on Developing Asia and Its Relevance to the Bank
110. 20	and Prices in Asian Rice Production		—E.M. Pernia, May 1987
	—William James and Teresita Ramirez, July 1983	No. 41	Emerging Issues in Asia and Social Cost
No. 21	Inflationary Effects of Exchange Rate		Benefit Analysis
	Changes in Nine Asian LDCs	NT 10	—I. Ali, September 1988
	—Pradumna B. Rana and J. Malcolm Dowling, Jr.,	No. 42	Shifting Revealed Comparative Advantage:
	December 1983		Experiences of Asian and Pacific Developing

	Countries —P.B. Rana, November 1988	No. 56	A Framework for Justifying Bank-Assisted Education Projects in Asia: A Review
No. 43	Agricultural Price Policy in Asia:		of the Socioeconomic Analysis
110. 10	Issues and Areas of Reforms		and Identification of Areas of Improvement
	—I. Ali, November 1988		-Etienne Van De Walle, February 1992
No. 44	Service Trade and Asian Developing Economies	No. 57	Medium-term Growth-Stabilization
-147	-M.G. Quibria, October 1989		Relationship in Asian Developing Countries
No. 45	A Review of the Economic Analysis of Power		and Some Policy Considerations
	Projects in Asia and Identification of Areas		-Yun-Hwan Kim, February 1993
	of Improvement	No. 58	Urbanization, Population Distribution,
	—I. Ali, November 1989		and Economic Development in Asia
No. 46	Growth Perspective and Challenges for Asia:		-Ernesto M. Pernia, February 1993
	Areas for Policy Review and Research	No. 59	The Need for Fiscal Consolidation in Nepal:
	—I. Ali, November 1989		The Results of a Simulation
No. 47	An Approach to Estimating the Poverty		-Filippo di Mauro and Ronald Antonio Butiong,
	Alleviation Impact of an Agricultural Project		July 1993
	—I. Ali, January 1990	No. 60	A Computable General Equilibrium Model
No. 48	Economic Growth Performance of Indonesia,		of Nepal
	the Philippines, and Thailand:		—Timothy Buehrer and Filippo di Mauro, October
	The Human Resource Dimension		1993
	—E.M. Pernia, January 1990	No. 61	The Role of Government in Export Expansion
No. 49	Foreign Exchange and Fiscal Impact of a Project:		in the Republic of Korea: A Revisit
	A Methodological Framework for Estimation		—Yun-Hwan Kim, February 1994
	—I. Ali, February 1990	No. 62	Rural Reforms, Structural Change,
No. 50	Public Investment Criteria: Financial		and Agricultural Growth in
	and Economic Internal Rates of Return		the People's Republic of China
	—I. Ali, April 1990		—Bo Lin, August 1994
No. 51	Evaluation of Water Supply Projects:	No. 63	Incentives and Regulation for Pollution Abatement
	An Economic Framework		with an Application to Waste Water Treatment
	—Arlene M. Tadle, June 1990		—Sudipto Mundle, U. Shankar, and Shekhar
No. 52	Interrelationship Between Shadow Prices, Project	37 04	Mehta, October 1995
	Investment, and Policy Reforms:	No. 64	Saving Transitions in Southeast Asia
	An Analytical Framework	37 05	—Frank Harrigan, February 1996
37 70	—I. Ali, November 1990	No. 65	Total Factor Productivity Growth in East Asia:
No. 53	Issues in Assessing the Impact of Project		A Critical Survey
	and Sector Adjustment Lending	NT 00	—Jesus Felipe, September 1997
NT 54	—I. Ali, December 1990	No. 66	Foreign Direct Investment in Pakistan:
No. 54	Some Aspects of Urbanization		Policy Issues and Operational Implications
	and the Environment in Southeast Asia		—Ashfaque H. Khan and Yun-Hwan Kim, July
No EF	—Ernesto M. Pernia, January 1991	No. 67	1999 Figoral Policy Income Distribution and Counth
No. 55	Financial Sector and Economic Development: A Survey	No. 67	Fiscal Policy, Income Distribution and Growth —Sailesh K. Jha, November 1999
	—Jungsoo Lee, September 1991		—5anesa A. Ina, wovember 1333
	— о индоио Lee, Бернениост 1331		

ECONOMIC STAFF PAPERS (ES)

No. 1	International Reserves:		—Ulrich Hiemenz, March 1982
	Factors Determining Needs and Adequacy	No. 8	Petrodollar Recycling 1973-1980.
	—Evelyn Go, May 1981		Part 1: Regional Adjustments and
No. 2	Domestic Savings in Selected Developing		the World Economy
	Asian Countries		—Burnham Campbell, April 1982
	—Basil Moore, assisted by A.H.M. Nuruddin	No. 9	Developing Asia: The Importance
	Chowdhury, September 1981		of Domestic Policies
No. 3	Changes in Consumption, Imports and Exports		-Economics Office Staff under the direction of Seiji
	of Oil Since 1973: A Preliminary Survey of		Naya, May 1982
	the Developing Member Countries	No. 10	Financial Development and Household
	of the Asian Development Bank		Savings: Issues in Domestic Resource
	—Dal Hyun Kim and Graham Abbott, September		Mobilization in Asian Developing Countries
	1981		-Wan-Soon Kim, July 1982
No. 4	By-Passed Areas, Regional Inequalities,	No. 11	Industrial Development: Role of Specialized
	and Development Policies in Selected		Financial Institutions
	Southeast Asian Countries		-Kedar N. Kohli, August 1982
	—William James, October 1981	No. 12	Petrodollar Recycling 1973-1980.
No. 5	Asian Agriculture and Economic Development		Part II: Debt Problems and an Evaluation
	-William James, March 1982		of Suggested Remedies
No. 6	Inflation in Developing Member Countries:		—Burnham Campbell, September 1982
	An Analysis of Recent Trends	No. 13	Credit Rationing, Rural Savings, and Financial
	—A.H.M. Nuruddin Chowdhury and J. Malcolm		Policy in Developing Countries
	Dowling, March 1982		-William James, September 1982
No. 7	Industrial Growth and Employment in	No. 14	Small and Medium-Scale Manufacturing
	Developing Asian Countries: Issues and		Establishments in ASEAN Countries:
	Perspectives for the Coming Decade		Perspectives and Policy Issues
	<u>-</u>		- •

	-Mathias Bruch and Ulrich Hiemenz, March 1983	No. 39	Domestic Adjustment to External Shocks
No. 15	Income Distribution and Economic		in Developing Asia
	Growth in Developing Asian Countries		—Jungsoo Lee, October 1987
	—J. Malcolm Dowling and David Soo, March 1983	No. 40	Improving Domestic Resource Mobilization
No. 16	Long-Run Debt-Servicing Capacity of		through Financial Development: Indonesia
	Asian Developing Countries: An Application		—Philip Erquiaga, November 1987
	of Critical Interest Rate Approach	No. 41	Recent Trends and Issues on Foreign Direct
	—Jungsoo Lee, June 1983	110. 11	Investment in Asian and Pacific Developing
No. 17	External Shocks, Energy Policy,		Countries
110. 11	and Macroeconomic Performance of Asian		—P.B. Rana, March 1988
	Developing Countries: A Policy Analysis	No. 42	Manufactured Exports from the Philippines:
	-William James, July 1983	110. 42	A Sector Profile and an Agenda for Reform
No. 10			
No. 18	The Impact of the Current Exchange Rate	NT- 40	—I. Ali, September 1988
	System on Trade and Inflation of Selected	No. 43	A Framework for Evaluating the Economic
	Developing Member Countries		Benefits of Power Projects
37 40	—Pradumna Rana, September 1983	37	—I. Ali, August 1989
No. 19	Asian Agriculture in Transition: Key Policy Issues	No. 44	Promotion of Manufactured Exports in Pakistan
	—William James, September 1983		—Jungsoo Lee and Yoshihiro Iwasaki, September
No. 20	The Transition to an Industrial Economy		1989
	in Monsoon Asia	No. 45	Education and Labor Markets in Indonesia:
	—Harry T. Oshima, October 1983		A Sector Survey
No. 21	The Significance of Off-Farm Employment		-Ernesto M. Pernia and David N. Wilson,
	and Incomes in Post-War East Asian Growth		September 1989
	—Harry T. Oshima, January 1984	No. 46	Industrial Technology Capabilities
No. 22	Income Distribution and Poverty in Selected	1101 10	and Policies in Selected ADCs
110. 22	Asian Countries		—Hiroshi Kakazu, June 1990
	—John Malcolm Dowling, Jr., November 1984	No. 47	Designing Strategies and Policies
M- 00	e, ,	10. 47	
No. 23	ASEAN Economies and ASEAN Economic		for Managing Structural Change in Asia
	Cooperation	37 40	—Ifzal Ali, June 1990
	—Narongchai Akrasanee, November 1984	No. 48	The Completion of the Single European Community
No. 24	Economic Analysis of Power Projects		Market in 1992: A Tentative Assessment of its
	—Nitin Desai, January 1985		Impact on Asian Developing Countries
No. 25	Exports and Economic Growth in the Asian Region		—J.P. Verbiest and Min Tang, June 1991
	—Pradumna Rana, February 1985	No. 49	Economic Analysis of Investment in Power Systems
No. 26	Patterns of External Financing of DMCs		—Ifzal Ali, June 1991
	—E. Go, May 1985	No. 50	External Finance and the Role of Multilateral
No. 27	Industrial Technology Development		Financial Institutions in South Asia:
	the Republic of Korea		Changing Patterns, Prospects, and Challenges
	—S.Y. Lo, July 1985		—Jungsoo Lee, November 1991
No. 28	Risk Analysis and Project Selection:	No. 51	The Gender and Poverty Nexus: Issues and
110. 20	A Review of Practical Issues	1101 01	Policies
	—J.K. Johnson, August 1985		-M.G. Quibria, November 1993
No. 29	Rice in Indonesia: Price Policy and Comparative	No. 52	The Role of the State in Economic Development:
110. 23		110. 52	Theory, the East Asian Experience,
	Advantage		· · · · · · · · · · · · · · · · · · ·
NT- 90	—I. Ali, January 1986		and the Malaysian Case
No. 30	Effects of Foreign Capital Inflows	37 50	—Jason Brown, December 1993
	on Developing Countries of Asia	No. 53	The Economic Benefits of Potable Water Supply
	—Jungsoo Lee, Pradumna B. Rana, and Yoshihiro		Projects to Households in Developing Countries
	Iwasaki, April 1986		—Dale Whittington and Venkateswarlu Swarna,
No. 31	Economic Analysis of the Environmental		January 1994
	Impacts of Development Projects	No. 54	Growth Triangles: Conceptual Issues
	—John A. Dixon et al., EAPI, East-West Center,		and Operational Problems
	August 1986		-Min Tang and Myo Thant, February 1994
No. 32	Science and Technology for Development:	No. 55	The Emerging Global Trading Environment
	Role of the Bank		and Developing Asia
	-Kedar N. Kohli and Ifzal Ali, November 1986		—Arvind Panagariya, M.G. Quibria, and Narhari
No. 33	Satellite Remote Sensing in the Asian		Rao, July 1996
1.0. 00	and Pacific Region	No. 56	Aspects of Urban Water and Sanitation in
	—Mohan Sundara Rajan, December 1986	110. 00	the Context of Rapid Urbanization in
No. 34	Changes in the Export Patterns of Asian and		Developing Asia
110. 54	Pacific Developing Countries: An Empirical		1 0
			—Ernesto M. Pernia and Stella LF. Alabastro,
	Overview 1007	NI. FF	September 1997
37 05	—Pradumna B. Rana, January 1987	No. 57	Challenges for Asia's Trade and Environment
No. 35	Agricultural Price Policy in Nepal	37	—Douglas H. Brooks, January 1998
	—Gerald C. Nelson, March 1987	No. 58	Economic Analysis of Health Sector Projects-
No. 36	Implications of Falling Primary Commodity		A Review of Issues, Methods, and Approaches
	Prices for Agricultural Strategy in the Philippines		-Ramesh Adhikari, Paul Gertler, and Anneli
	—Ifzal Ali, September 1987		Lagman, March 1999
3.T O.F		No. 59	The Asian Crisis: An Alternate View
No. 37	Determining Irrigation Charges: A Framework	110. 00	The Holdin Cholo, and Internate view
No. 37	Determining Irrigation Charges: A Framework —Prabhakar B. Ghate, October 1987	110. 00	—Rajiv Kumar and Bibek Debroy, July 1999
No. 37		No. 60	
	—Prabhakar B. Ghate, October 1987		—Rajiv Kumar and Bibek Debroy, July 1999
	—Prabhakar B. Ghate, October 1987 The Role of Fertilizer Subsidies in Agricultural Production: A Review of Select Issues		—Rajiv Kumar and Bibek Debroy, July 1999 Social Consequences of the Financial Crisis in Asia
	—Prabhakar B. Ghate, October 1987 The Role of Fertilizer Subsidies in Agricultural		—Rajiv Kumar and Bibek Debroy, July 1999 Social Consequences of the Financial Crisis in

OCCASIONAL PAPERS (OP)				
No. 1	Poverty in the People's Republic of China: Recent Developments and Scope for Bank Assistance	No. 12	Managing Development through Institution Building — <i>Hilton L. Root, October 1995</i>	
No. 2	—K.H. Moinuddin, November 1992 The Eastern Islands of Indonesia: An Overview of Development Needs and Potential	No. 13	Growth, Structural Change, and Optimal Poverty Interventions —Shiladitya Chatterjee, November 1995	
No. 3	—Brien K. Parkinson, January 1993 Rural Institutional Finance in Bangladesh and Nepal: Review and Agenda for Reforms	No. 14	Private Investment and Macroeconomic Environment in the South Pacific Island Countries: A Cross-Country Analysis	
No. 4	—A.H.M.N. Chowdhury and Marcelia C. Garcia, November 1993 Fiscal Deficits and Current Account Imbalances	No. 15	—T.K. Jayaraman, October 1996 The Rural-Urban Transition in Viet Nam: Some Selected Issues	
	of the South Pacific Countries: A Case Study of Vanuatu —T.K. Jayaraman, December 1993	No. 16	—Sudipto Mundle and Brian Van Arkadie, October 1997 A New Approach to Setting the Future	
No. 5 No. 6	Reforms in the Transitional Economies of Asia —Pradumna B. Rana, December 1993 Environmental Challenges in the People's Republic		Transport Agenda —Roger Allport, Geoff Key, and Charles Melhuish, June 1998	
	of China and Scope for Bank Assistance —Elisabetta Capannelli and Omkar L. Shrestha, December 1993	No. 17	Adjustment and Distribution: The Indian Experience —Sudipto Mundle and V.B. Tulasidhar, June 1998	
No. 7	Sustainable Development Environment and Poverty Nexus —K.F. Jalal, December 1993	No. 18 No. 19	Tax Reforms in Viet Nam: A Selective Analysis —Sudipto Mundle, December 1998 Surges and Volatility of Private Capital Flows to	
No. 8	Intermediate Services and Economic Development: The Malaysian Example —Sutanu Behuria and Rahul Khullar, May 1994		Asian Developing Countries: Implications for Multilateral Development Banks —Pradumna B. Rana, December 1998	
No. 9	Interest Rate Deregulation: A Brief Survey of the Policy Issues and the Asian Experience —Carlos J. Glower, July 1994	No. 20	The Millennium Round and the Asian Economies: An Introduction —Dilip K. Das, October 1999	
No. 10	Some Aspects of Land Administration in Indonesia: Implications for Bank Operations —Sutanu Behuria, July 1994	No. 21	Occupational Segregation and the Gender Earnings Gap —Joseph E. Zveglich, Jr. and Yana van der Meulen	
No. 11	Demographic and Socioeconomic Determinants of Contraceptive Use among Urban Women in the Melanesian Countries in the South Pacific: A Case Study of Port Vila Town in Vanuatu —T.K. Jayaraman, February 1995	No. 22	Rodgers, December 1999 Information Technology: Next Locomotive of Growth? —Dilip K. Das, June 2000	
	STATISTICAL REI	PORT SEI	RIES (SR)	
No. 1	Estimates of the Total External Debt of the Developing Member Countries of ADB: 1981-1983	No. 10	in Asian Developing Countries, 1985 —Jungsoo Lee and I.P. David, April 1987 A Survey of the External Debt Situation	
No. 2	—I.P. David, September 1984 Multivariate Statistical and Graphical Classification Techniques Applied to the Problem of Grouping Countries	No. 11	in Asian Developing Countries, 1986 —Jungsoo Lee and I.P. David, April 1988 Changing Pattern of Financial Flows to Asian and Pacific Developing Countries	

No. 1	Estimates of the Total External Debt of the Developing Member Countries of ADB: 1981-1983 —I.P. David, September 1984 Multivariate Statistical and Graphical	No. 10	in Asian Developing Countries, 1985 —Jungsoo Lee and I.P. David, April 1987 A Survey of the External Debt Situation in Asian Developing Countries, 1986
N0. Z	Classification Techniques Applied to the Problem of Grouping Countries —I.P. David and D.S. Maligalig, March 1985	No. 11	—Jungsoo Lee and I.P. David, April 1988 Changing Pattern of Financial Flows to Asian and Pacific Developing Countries —Jungsoo Lee and I.P. David, March 1989
No. 3	Gross National Product (GNP) Measurement Issues in South Pacific Developing Member Countries of ADB —S.G. Tiwari, September 1985	No. 12	The State of Agricultural Statistics in Southeast Asia —I.P. David, March 1989
No. 4	Estimates of Comparable Savings in Selected DMCs —Hananto Sigit, December 1985	No. 13	A Survey of the External Debt Situation in Asian and Pacific Developing Countries: 1987-1988 —Jungsoo Lee and I.P. David, July 1989
No. 5	Keeping Sample Survey Design and Analysis Simple —I.P. David, December 1985	No. 14	A Survey of the External Debt Situation in Asian and Pacific Developing Countries: 1988-1989 —Jungsoo Lee, May 1990
No. 6	External Debt Situation in Asian Developing Countries —I.P. David and Jungsoo Lee, March 1986	No. 15	A Survey of the External Debt Situation in Asian and Pacific Developing Countries: 1989- 1992
No. 7	Study of GNP Measurement Issues in the South Pacific Developing Member Countries. Part I: Existing National Accounts of SPDMCs-Analysis of Methodology and Application of SNA Concepts —P. Hodgkinson, October 1986	No. 16	-Min Tang, June 1991 Recent Trends and Prospects of External Debt Situation and Financial Flows to Asian and Pacific Developing Countries -Min Tang and Aludia Pardo, June 1992
No. 8	Study of GNP Measurement Issues in the South Pacific Developing Member Countries. Part II: Factors Affecting Intercountry Comparability of Per Capita GNP —P. Hodgkinson, October 1986	No. 17	Purchasing Power Parity in Asian Developing Countries: A Co-Integration Test —Min Tang and Ronald Q. Butiong, April 1994 Capital Flows to Asian and Pacific Developing Countries: Recent Trends and Future Prospects —Min Tang and James Villafuerte, October 1995
No. 9	Survey of the External Debt Situation		—min lang and sames vitaluerie, October 1995

SERIALS

(Available commercially through ADB Office of External Relations)

- 1. Asian Development Outlook (ADO; annual) \$36.00 (paperback)
- 2. Key Indicators of Developing Asian and Pacific Countries (KI; annual) \$35.00 (paperback)
- 3. Asian Development Review (ADR; semiannual) \$5.00 per issue; \$8.00 per year (2 issues)

SPECIAL STUDIES, CO-PUBLISHED

(Available commercially through Oxford University Press Offices, Edward Elgar Publishing, and Palgrave MacMillan)

FROM OXFORD UNIVERSITY PRESS: Oxford University Press (China) Ltd 18th Floor, Warwick House East Taikoo Place, 979 King's Road Quarry Bay, Hong Kong Tel (852) 2516 3222 Fax (852) 2565 8491

E-mail: webmaster@oupchina.com.hkWeb: www.oupchina.com.hk

\$25.00 (paperback)

- Informal Finance: Some Findings from Asia Prabhu Ghate et. al., 1992 \$15.00 (paperback)
- Mongolia: A Centrally Planned Economy in Transition
 Asian Development Bank, 1992
 \$15.00 (paperback)
- Rural Poverty in Asia, Priority Issues and Policy Options
 Edited by M.G. Quibria, 1994
- Growth Triangles in Asia: A New Approach to Regional Economic Cooperation
 Edited by Myo Thant, Min Tang, and Hiroshi Kakazu 1st ed., 1994 \$36.00 (hardbound)

 Revised ed., 1998 \$55.00 (hardbound)
- 5. Urban Poverty in Asia: A Survey of Critical Issues Edited by Ernesto Pernia, 1994
 \$18.00 (paperback)
- 6. Critical Issues in Asian Development:
 Theories, Experiences, and Policies
 Edited by M.G. Quibria, 1995
 \$15.00 (paperback)
 \$36.00 (hardbound)
- Financial Sector Development in Asia Edited by Shahid N. Zahid, 1995 \$50.00 (hardbound)
- Financial Sector Development in Asia: Country Studies *Edited by Shahid N. Zahid*, 1995 \$55.00 (hardbound)
- 9. Fiscal Management and Economic Reform in the People's Republic of China Christine P.W. Wong, Christopher Heady, and Wing T. Woo, 1995 \$15.00 (paperback)
- From Centrally Planned to Market Economies:
 The Asian Approach
 Edited by Pradumna B. Rana and Naved Hamid, 1995

Vol. 1: Overview \$36.00 (hardbound)

Vol. 2: People's Republic of China and Mongolia $\$50.00\ (hardbound)$

Vol. 3: Lao PDR, Myanmar, and Viet Nam \$50.00 (hardbound)

- Current Issues in Economic Development:
 An Asian Perspective
 Edited by M.G. Quibria and J. Malcolm Dowling, 1996
 \$50.00 (hardbound)
- The Bangladesh Economy in Transition *Edited by M.G. Quibria*, 1997 \$20.00 (hardbound)
- The Global Trading System and Developing Asia Edited by Arvind Panagariya, M.G. Quibria, and Narhari Rao, 1997 \$55.00 (hardbound)
- Social Sector Issues in Transitional Economies of Asia *Edited by Douglas H. Brooks and Myo Thant, 1998* \$25.00 (paperback) \$55.00 (hardbound)
- Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future Edited by Yun-Hwan Kim and Paul Smoke, 2003 \$15.00 (paperback)
- Local Government Finance and Bond Markets *Edited by Yun-Hwan Kim*, 2003 \$15.00 (paperback)

FROM EDWARD ELGAR:

Marston Book Services Limited PO Box 269, Abingdon Oxon OX14 4YN, United Kingdom Tel +44 1235 465500 Fax +44 1235 465555 Email: direct.order@marston.co.uk Web: www.marston.co.uk

 Reducing Poverty in Asia: Emerging Issues in Growth, Targeting, and Measurement Edited by Christopher M. Edmonds, 2003

FROM PALGRAVE MACMILLAN:

Palgrave Macmillan Ltd Houndmills, Basingstoke Hampshire RG21 6XS, United Kingdom Tel: +44 (0)1256 329242 Fax: +44 (0)1256 479476 Email: orders@palgrave.com Web: www.palgrave.com/home/

Managing FDI in a Globalizing Economy
 Asian Experiences
 Edited by Douglas H. Brooks and Hal Hill, 2004

2. Poverty, Growth, and Institutions in Developing Asia Edited by Ernesto M. Pernia and Anil B. Deolalikar, 2003

SPECIAL STUDIES, IN-HOUSE

(Available commercially through ADB Office of External Relations)

- Rural Poverty in Developing Asia *Edited by M.G. Quibria*
 - Vol. 1: Bangladesh, India, and Sri Lanka, 1994 \$35.00 (paperback)
 - Vol. 2: Indonesia, Republic of Korea, Philippines, and Thailand, 1996 \$35.00 (paperback)
- 2. Gender Indicators of Developing Asian and Pacific Countries
 - Asian Development Bank, 1993 \$25.00 (paperback)
- External Shocks and Policy Adjustments: Lessons from the Gulf Crisis
 Edited by Naved Hamid and Shahid N. Zahid, 1995
- \$15.00 (paperback)
 4. Indonesia-Malaysia-Thailand Growth Triangle:
 Theory to Practice
 Edited by Myo Thant and Min Tang, 1996
 \$15.00 (paperback)
- Emerging Asia: Changes and Challenges Asian Development Bank, 1997 \$30.00 (paperback)
- 6. Asian Exports
 Edited by Dilip Das, 1999
 \$35.00 (paperback)
 \$55.00 (hardbound)
- 7. Development of Environment Statistics in Developing Asian and Pacific Countries Asian Development Bank, 1999 \$30.00 (paperback)
- 8. Mortgage-Backed Securities Markets in Asia
 Edited by S.Ghon Rhee & Yutaka Shimomoto, 1999
 \$35.00 (paperback)
- 9. Rising to the Challenge in Asia: A Study of Financial Markets

Asian Development Bank

- Vol. 1: An Overview, 2000 \$20.00 (paperback)
- Vol. 2: Special Issues, 1999 \$15.00 (paperback)
- Vol. 3: Sound Practices, 2000 \$25.00 (paperback)
- Vol. 4: People's Republic of China, 1999 \$20.00 (paperback)
- Vol. 5: India, 1999 \$30.00 (paperback)
- Vol. 6: Indonesia, 1999 \$30.00 (paperback)
- Vol. 7: Republic of Korea, 1999 \$30.00 (paperback)
- Vol. 8: Malaysia, 1999 \$20.00 (paperback)
- Vol. 9: Pakistan, 1999 \$30.00 (paperback)
- Vol. 10: Philippines, 1999 \$30.00 (paperback)
- Vol. 11: Thailand, 1999 \$30.00 (paperback)
- Vol. 12: Socialist Republic of Viet Nam, 1999 \$30.00 (paperback)
- Corporate Governance and Finance in East Asia:
 A Study of Indonesia, Republic of Korea, Malaysia, Philippines and Thailand
 J. Zhuang, David Edwards, D. Webb, & Ma. Virginita Capulong

- Vol. 1: A Consolidated Report, 2000 \$10.00 (paperback) Vol. 2: Country Studies, 2001 \$15.00 (paperback)
- 11. Financial Management and Governance Issues

Asian Development Bank, 2000

Cambodia \$10.00 (paperback)

People's Republic of China \$10.00 (paperback)

Mongolia \$10.00 (paperback)

Pakistan \$10.00 (paperback)

Papua New Guinea \$10.00 (paperback)

Uzbekistan \$10.00 (paperback)

Viet Nam \$10.00 (paperback)

Selected Developing Member Countries \$10.00 (paperback)

- 12. Government Bond Market Development in Asia

 Edited by Yun-Hwan Kim, 2001

 \$25.00 (paperback)
- 13. Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future

 Edited by Paul Smoke and Yun-Hwan Kim, 2002

 \$15.00 (paperback)
- Guidelines for the Economic Analysis of Projects Asian Development Bank, 1997 \$10.00 (paperback)
- Guidelines for the Economic Analysis of Telecommunications Projects Asian Development Bank, 1997 \$10.00 (paperback)
- Handbook for the Economic Analysis of Water Supply Projects Asian Development Bank, 1999 \$10.00 (hardbound)
- 17. Handbook for the Economic Analysis of Health Sector Projects
 Asian Development Bank, 2000
 \$10.00 (paperback)
- Handbook for Integrating Povery Impact Assessment in the Economic Analysis of Projects Asian Development Bank, 2001 \$10.00 (paperback)
- Handbook for Integrating Risk Analysis in the Economic Analysis of Projects
 Asian Development Bank, 2002
 \$10.00 (paperback)
- 20. Handbook on Environment Statistics
 Asian Development Bank, 2002
 \$10.00 (hardback)
- Defining an Agenda for Poverty Reduction, Volume 1
 Edited by Christopher Edmonds and Sara Medina, 2002
 \$15.00 (paperback)
- Defining an Agenda for Poverty Reduction, Volume 2
 Edited by Isabel Ortiz, 2002
 \$15.00 (paperback)
- Economic Analysis of Policy-based Operations: Key Dimensions
 Asian Development Bank, 2003
 \$10.00 (paperback)

- Myers, S., and N. Majluf, 1984. "Corporate Financing Decision When Firms Have Information Investors Do Not Have." Journal of Financial Economics 13:187-222.
- OECD, 2000. *Directions of Recent Pension Reforms in OECD Countries*. Organization of Economic Cooperation and Development, Paris.
- Pistor, K., Y. Keinan, J. Kleinheisterkamp, and M. West, 2003. "Evolution of Corporate Law and the Transplant Effect: Lessons from Six Countries." *The World Bank Research Observer* 18(1):89-112.
- Rajan, R. and L. Zingales, 2001. "Financial Systems, Industrial Structure and Growth." Oxford Review of Economic Policy 17(Winter, 4):467-82.
- ———, 2003. "The Great Reversal: The Politics of Financial Development in the 20th Century." *Journal of Financial Economics*. Forthcoming.
- Rodrik, D., A. Subramanian, and F. Trebbi, 2002, Institutions Rule: The Primacy of Institutions Over Geography and Integration in Economic Development. NBER Working Paper 9305, National Bureau of Economic Research, Massachusetts.
- Roe, M. J., 1994. Strong Managers, Weak Owners: The Political Roots of American Corporate Finance. Princeton: Princeton University Press.
- Shirai, S., 2001, Overview of Financial Markets in Asia¾Cases of the Republic of Korea, Malaysia, Thailand and Indonesia. ADBI Working Paper 25, Asian Development Bank Institute, Tokyo.
- Stone, M., 2000. The Corporate Sector Dynamics of Systemic Financial Crisis. IMF Working Paper WP/00/114, International Monetary Fund, Washington, DC.
- Summers, L. H., 1999. "Reflections on Managing Global Integration." *Journal of Economic Perspectives* 13:3-18.
- Sylla, R., 1995. The Rise of Security Markets: What Can Government Do? WPS 1539, World Bank, Washington, DC.
- Turner, P., 2002. Bond Markets in Emerging Economies: An Overview of Policy Issues. In BIS Papers No. 11, The Development of Bond Markets in Emerging Economies, Monetary and Economic Department, Bank for International Settlements,
- Tornell, A., 2002. "Policy in an Economy with Balance Sheet Effects." In S. Edwards and J. A. Frankel, eds., Preventing Currency Crisis in Emerging Markets. Chicago, IL: University of Chicago Press.